Author's response to reviews

Title: Parasitic dermoid cyst managed laparoscopically: a case report

Authors:

Ciprian Bartlett (infiniteworx@hotmail.com)
Ashfaq Khan (Ashfaq.Khan@whittington.nhs.uk)
Narendra Pisal (pisal@dr.com)

Version: 2 Date: 17 May 2008

Author's response to reviews:

17 May 2008

The Editorial team
Journal of Medical Case Reports

As per your request the following modifications have been made to the abstract and case presentation. The abbreviation POD in the abstract has been replaced by Pouch of Douglas and the patient’s ethnic origin has been entered in both the abstract and the case presentation.

Yours sincerely,

CE Bartlett MRCOG,